Author's response to reviews

**Title:** Novel Variants in the PRDX6 Gene and the Risk of Acute Lung Injury Following Major Trauma

**Authors:**

Melanie Rushefski (meldoran@mail.med.upenn.edu)
Richard Aplenc (raplenc@mail.med.upenn.edu)
Nuala Meyer (Nuala.Meyer@uphs.upenn.edu)
Mingyao Li (mingyao@mail.med.upenn.edu)
Rui Feng (ruifeng@mail.med.upenn.edu)
Paul N. Lanken (Paul.Lanken@uphs.upenn.edu)
Robert Gallop (rgallop@wcupa.edu)
Scarlett Bellamy (bellamys@mail.med.upenn.edu)
A. Russell Localio (rlocalio@mail.med.upenn.edu)
Sheldon I. Feinstein (sif@mail.med.upenn.edu)
Aron B. Fisher (abf@exchange.upenn.edu)
Steven M. Albelda (albelda@mail.med.upenn.edu)
Jason D. Christie (jchristi@mail.med.upenn.edu)

**Version:** 6  **Date:** 17 May 2011

**Author's response to reviews:** see over
Author Response:

We have reviewed our manuscript once more and made a few minor adjustments. After making these changes, we feel our manuscript is acceptable for publication.

Thank you.